

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Gietl, Daniel; Haufler, Andreas

Working Paper Bonus Taxes and International Competition for Bank Managers

CESifo Working Paper, No. 6495

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Gietl, Daniel; Haufler, Andreas (2017) : Bonus Taxes and International Competition for Bank Managers, CESifo Working Paper, No. 6495, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: https://hdl.handle.net/10419/167481

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Bonus Taxes and International Competition for Bank Managers

Daniel Gietl, Andreas Haufler



Impressum:

CESifo Working Papers ISSN 2364-1428 (electronic version) Publisher and distributor: Munich Society for the Promotion of Economic Research - CESifo GmbH The international platform of Ludwigs-Maximilians University's Center for Economic Studies and the ifo Institute Poschingerstr. 5, 81679 Munich, Germany Telephone +49 (0)89 2180-2740, Telefax +49 (0)89 2180-17845, email <u>office@cesifo.de</u> Editors: Clemens Fuest, Oliver Falck, Jasmin Gröschl www.cesifo-group.org/wp

An electronic version of the paper may be downloaded

- · from the SSRN website: <u>www.SSRN.com</u>
- from the RePEc website: <u>www.RePEc.org</u>
- from the CESifo website: <u>www.CESifo-group.org/wp</u>

Bonus Taxes and International Competition for Bank Managers

Abstract

We analyze the competition in bonus taxation when banks compensate their managers by means of fixed and incentive pay and bankers are internationally mobile. Banks choose bonus payments that induce excessive managerial risk-taking to maximize their private benefits of existing government bailout guarantees. In this setting the international competition in bonus taxes may feature a 'race to the bottom' or a 'race to the top', depending on whether bankers are a source of net positive tax revenue or inflict net fiscal losses on taxpayers as a result of incentive pay. A 'race to the top' becomes more likely when governments' impose only lax capital requirements on banks, whereas a 'race to the bottom' is more likely when bank losses are partly collectivized in a banking union.

JEL-Codes: H200, H870, G280.

Keywords: bonus taxes, international tax competiton, migration.

Daniel Gietl Seminar for Economic Policy University of Munich Akedemiestr. 1/II Germany – 80799 Munich Daniel.Gietl@econ.lmu.de Andreas Haufler Seminar for Economic Policy University of Munich Akedemiestr. 1/II Germany – 80799 Munich Andreas.Haufler@econ.lmu.de

May 2017

Paper presented at conferences and seminars in Augsburg (German Economic Association), Berlin, Glasgow (Strathclyde), Lake Tahoe (IIPF), Munich (CESifo), Oxford, and Rostock.We thank Hendrik Hakenes, Christian Holzner, Kai Konrad, Michael Kriebel, Dominika Langenmayr, Ben Lockwood, Colin von Negenborn, Doris Neuberger, Dirk Schindler, Davide Suverato and Ian Wooton for helpful comments. Financial support from the German Research Foundation through CRC TRR 190 and Grant No. HA 3195/9-1 is gratefully acknowledged.

1 Introduction

Bankers' bonuses have been the cause of much debate, and resentment, in recent years. Steep incentive schemes for bank managers have been identified as one of the root causes for the global financial crisis of 2008, as bonuses are believed to be responsible for excessive risk-taking in the banking sector.¹ Empirical evidence confirms that incentive pay has been positively correlated with risk-taking in the pre-crisis period (e.g. Bhagat and Bolton, 2014; Efing et al., 2015). In addition, bankers' bonuses play a significant role in the rising inequality of incomes in many developed countries. Bell and Van Reenen (2014) estimate, for example, that rising bonuses paid to bankers account for two-thirds of the increase in the share of the top 1% of the income distribution in the United Kingdom since 1999. For the United States, Philippon and Reshef (2012) find that, from the mid-1990s to 2006, chief executive officers (CEOs) in the finance industry have earned a 250% premium relative to CEOs in other sectors of the economy.

In response to these developments, several countries have introduced bonus taxes. The UK introduced a one-time 50% withholding tax on banker bonuses that exceeded GBP 25,000 and were paid between December 2009 and April 2010.² France followed with a similar, temporary bonus tax of 50% in 2010. In the United States, the House of Representatives approved, in 2009, a 90% bonus tax for traders, executives and bankers working for companies holding at least \$ 5 billion in bailout money and earning more than \$ 250,000 in family income. This bill was blocked in the U.S. Senate, however. Since 2010, Italy levies a permanent, 10% additional bonus tax for the banking sector, if variable compensation exceeds three times the fixed salary. In parallel to these national bonus taxes, the European Union has introduced, as of 2014, a regulation that limits bonuses paid to high-level managers in the financial sector to 100% of their fixed salary (200% with shareholder approval).

Given the massive side effects of bankers' bonuses and the strong public sentiment against them it is surprising, however, that bonus taxation has not become more common, or more persistent. One critical argument for why bankers' bonuses are not taxed

¹See e.g. Alan Blinder's commentary "Crazy compensation and the crisis", The Wall Street Journal, 28 May 2009.

²The UK bonus tax has been empirically analyzed by Ehrlich and Radulescu (2017). The authors find that the introduction of the bonus tax has led to a 40% fall in bonus payments. However, other components of executive pay have simultaneously been raised so as to largely compensate bank managers for the reduction in their bonuses.

more is that top bankers might leave a country that taxes their bonuses severely, and work instead for a bank abroad. Indeed, there is ample evidence that bank managers are mobile across countries. The largest German bank, the Deutsche Bank, for example, has been consecutively governed by three non-German CEOs since 2002.³ More generally, there is a substantial literature indicating that the international mobility of top managers has grown substantially over the past two decades (e.g. Greve et al., 2015). Focusing specifically on the finance industry, Greve et al. (2009) investigate the nationality of board executives in 41 large European firms in the banking and insurance industry and find that 26% of all executives in the sample are non-nationals. Similarly, Staples (2008) finds that among the 48 largest commercial banks in the world close to 70% have at least one non-national board member.

Despite the conclusive evidence for the international mobility of bank managers, almost all theoretical papers investigating the impact of banker bonus taxation and regulation use a closed-economy framework (see our literature review below). In this paper we aim to fill this gap by analyzing the non-cooperative setting of bonus taxes in a twocountry model with one bank in each country and mobility of bankers between the two banks. Our model incorporates governments, banks, and bank managers that all behave optimally, given their incentives. The model has four stages. In the first stage, the two symmetric countries non-cooperatively set bonus taxes to maximize expected net revenues, which result as the difference between expected bonus tax revenues and the expected costs to taxpayers of bailing out the bank in the case of default. In the second stage, the two banks endogenously choose the managers' compensation structure, which consists of bonus payments and a fixed wage component. The contracts set in this stage take simultaneous effort and risk-taking decisions in the country in which they work.

At the core of our analysis are two principal-agent problems. The first is between a bank's shareholders and its managers. Managers have private effort and risk-taking costs and thus choose lower effort and less risk-taking than would be optimal for shareholders. Second, there is a principal-agent problem between the banks' shareholders and taxpayers in the bank's home country, if shareholders anticipate that their bank is (partly) bailed out by the government in case of failure. Therefore, shareholders incen-

³Josef Ackermann (Switzerland) chaired the Deutsche Bank from 2002 to 2012. From 2012 until 2015, Anshuman Jain (UK) chaired the Deutsche Bank, together with Jürgen Fitschen. Since 2015, John Cryan (UK) is the Deutsche Bank's chief executive.

tivize bank managers to take on "excessive" risk, relative to what would be optimal for the country as a whole. Governments therefore choose bonus taxes for a double reason, to collect tax revenues and to make bonuses a more costly instrument from the bank's perspective. Both banks and governments compete with their respective counterparts in the foreign country.

Our main result is that there can be either a 'race to the bottom' or a 'race to the top' with respect to the bonus taxes chosen in the non-cooperative tax equilibrium. Which result is obtained depends on the fiscal value of a bank manager, which equals the expected bonus tax income minus the expected bailout costs for the government. A 'race to the top' is more likely to occur if the risks of bank failures are large, and if taxpayers are heavily exposed to downside risks as a result of low capital requirements for banks. In this case governments regard each banker as a fiscal liability and optimally set bonus taxes in excess of those that are globally optimal, on order to shift these risks from domestic to foreign taxpayers. A 'race to the top' becomes less likely, however, when bank profits also enter the government's objective function or when bailout costs for banks are collectivized. The latter occurs, for example, in the European Union's newly established banking union. Together these results may explain why several countries levied high bonus tax rates in the immediate aftermath of the financial crises, but abolished these taxes later, as the perceived risks to taxpayers fell while bank profits resumed.

Our analysis is related to two strands in the literature. A first strand analyzes the effects of public policies towards bonus schemes.⁴ Besley and Ghatak (2013) analyze the optimal bonus taxation of managers when bankers can choose both effort and risk-taking. Hakenes and Schnabel (2014) study how bailout expectations affect both the optimal bonus contract offered by the bank and the imposition of bonus caps by welfare-maximizing governments. Thanassoulis (2012) derives the role of bonus caps in a setting where the competition for bankers increases their compensation, which in turn drives up banks' default risk. All these studies analyze policies towards bonus pay in a closed economy setting. Radulescu (2012) is the only study of bonus taxation in an open economy of which we are aware. This paper does not incorporate risk-taking decisions by bank managers, however, and bonus taxes are exogenous to the model.

⁴The incentive effects of bonus schemes are themselves the subject of a large literature. See e.g. Bannier et al. (2013) and Acharya et al. (2016) for recent analyses of bonus pay in the competition for managerial talent.

A second related literature analyzes policy competition in the presence of cross-country externalities. There is an established literature on international tax competition (see Keen and Konrad, 2013, for a recent survey) that has recently been applied to study non-linear income tax competition in the presence of mobile high-income earners (Simula and Trannoy, 2010; Bierbrauer et al., 2013; Lehmann et al., 2014; Lipatov and Weichenrieder, 2015). These models generally find a 'race to the bottom' in income taxation, as a result of policy competition for mobile individuals.⁵ However, the mobile rich take no risks in these models, and they are always a source of positive tax revenue for the competing governments. As we show in this paper, the direction of tax competition may change when the competition is for bank managers, who may inflict fiscal losses on their home governments through overly risky investment choices.

This paper is structured as follows. Section 2 introduces the basic setup of our analysis. Section 3 analyzes the decisions of bank managers. Section 4 turns to the banks' choice of optimal managerial compensation. Section 5 derives the tax competition equilibrium between the two governments. Section 6 analyzes several model extensions and Section 7 concludes.

2 Model setup and roadmap

We consider a model of two banks, one in each of two symmetric countries $i \in (1, 2)$. Banks are run by risk-neutral shareholders. The size of each bank is variable, corresponding to the number of identical divisions within the bank. Running a division requires the specific knowledge of a bank manager. Hence each bank employs exactly one manager per division and the number of managers a bank hires equals the number of its divisions.

The total number of managers in our regional economy is fixed, and all managers are employed in one of the two countries in equilibrium. Managers differ in their individual attachment to the two countries and therefore are imperfectly mobile between the two countries. Apart from their location preferences, all managers are identical. Banks compete for the imperfectly mobile managers by means of a compensation package, which consists of both a fixed wage and a bonus payment. Managerial compensation

⁵Similar 'race to the bottom' results have been obtained for regulatory policies towards profitable banks that export part of their services to the foreign country. See e.g. Acharya (2003), Sinn (2003) and Dell'Ariccia and Marquez (2006).

is chosen so as to maximize the bank's after-tax profits, which is the product of the number of divisions within the bank, and the net expected profit per division.⁶

The bank's portfolio: Each division of a bank in country *i* has a total amount of fixed assets equal to one, which is lent in the world market. While the two banks in our model are large players in the regional market for managers, they are small investors in the world capital market. Hence the gross returns from investment are exogenous from each bank's perspective. Lending operations are risky. We assume that there are three possible returns for the bank, which can be high, medium, or low $(Y^h,$ Y^m and Y^l). The portfolio realizes a high return Y^h with probability $p^h > 0$, a medium return $Y^m < Y^h$ with probability $p^m > 0$ and a low return $Y^l = 0$ with probability $p^l = 1 - p^h - p^m > 0$. While the portfolio returns are exogenous and observable, the corresponding probabilities are endogenously determined by the unobservable decisions of managers to exert effort and take risk.⁷

Specifically, we assume that the probabilities for the different returns are linear functions of the manager's effort e and risk-taking choice r:

$$p^{h} = \alpha e + \beta r,$$

$$p^{m} = p_{0}^{m} - r,$$

$$p^{l} = p_{0}^{l} - \alpha e + (1 - \beta)r,$$
(1)

where $p^h + p^m + p^l = p_0^m + p_0^l \equiv 1$. The exogenous 'baseline' probability of a low state, p_0^l , can be interpreted as reflecting general business conditions in the banking sector. With the specification (1), a high return Y^h can only be obtained when managers either exert effort or take risk. More generally, the manager's effort *e* shifts probability mass from p^l to p^h and therefore increases the mean return of the division's portfolio. Risk-taking *r* instead shifts probability mass from the intermediate state to the high and the low states, thereby increasing the variance of the division's returns.

The managers' remuneration structure: To align the interests of each division manager and the bank's shareholders, the bank in country i pays a bonus, if the return

⁶An alternative setting would have national banking sectors that consist of a variable number of identical, small banks. In such a setting it is difficult, however, to model a meaningful principal-agent problem between the owner and the manager of a bank.

⁷This specification of an individual division's return structure follows Besley and Ghatak (2013).

for the bank division is high (Y^h) . Beyond the bonus payment, the managers also receive an endogenous wage that is paid independently of the realized return. The bonus payment will induce bank managers to increase both effort and risk-taking, relative to a situation where they receive only the fixed pay. The higher risk-taking will, however, also increase the probability that the low return $Y^l = 0$ occurs. In this case the division fails. For analytical tractability, we assume that the returns of the different divisions of a bank are perfectly correlated. Hence, if one division of a bank fails, so do all the others. Therefore, the bank as a whole fails with probability $p^{l.8}$

If the bank in country i fails, the external creditors of the bank will be bailed out by country i's government. We take the bank's external funds to be savings deposits and assume that the share of deposit financing is exogenously fixed, for example by a binding minimum capital requirement. In this case the bailout occurs through savings deposit insurance, which exists in virtually all developed countries.⁹ With these government guarantees, the bank does not face the full cost of failure. It therefore has an incentive to induce excessive risk-taking by its managers, as compared to the social optimum.¹⁰

By choosing their optimal compensation scheme, banks simultaneously influence the effort and risk-taking choices of their managers, and try to attract managers from abroad by offering an attractive overall level of managerial pay. Since effort and risk-taking choices can only be affected by the bonus, this instrument will always be part of the optimal compensation package. Whether the fixed wage will also be used in the bank's optimum then depends on which of the two compensation elements is more cost effective in attracting additional managers from abroad. Our analysis will therefore distinguish two regimes, depending on whether only the bonus is used in the bank's optimum (Regime I), or whether it is optimal for the bank to combine bonus payments and a fixed wage component (Regime II).

⁸If the returns of different divisions were imperfectly correlated, cross-subsidization between divisions within the bank would be possible. In this case a failure for the entire bank would still arise with a positive probability, but this probability would be a complex function of the correlation coefficient, the number of divisions, and the profitability of each division.

⁹The main argument in favor of deposit insurance is that it prevents bank-runs and thus stabilizes the banking system (Diamond and Dybvig, 1983). Barth et al. (2006) give an overview of deposit insurance schemes around the world, and discuss their benefits and costs.

¹⁰This argument also makes clear why the bank pays a bonus only in the high state. A bonus in the intermediate state m would not change effort and decrease risk-taking, thereby reducing the amount of loss-shifting to taxpayers [see eq. (1)]. Hence it would tend to reduce expected profits from the bank's perspective.

The government: Governments are aware of the moral hazard problem caused by their bailout policies and use bonus taxation to counteract the distorted incentives. Bonus taxes therefore have a corrective role in our setting, in addition to their objective of raising government revenue. In our benchmark model, we assume that the government maximizes its net tax revenue, which is given by total expected bonus tax revenue minus the expected bailout costs. This focus on tax revenue maximization corresponds to the declared objective of many governments to collect a 'fair and substantial contribution' from the financial sector for the fiscal cost this sector has caused during the financial crisis (International Monetary Fund, 2010).¹¹ More generally, as long as the value of government tax revenues exceeds that of private sector incomes, a bonus tax dominates the alternative policy instrument of a regulatory cap on bonus payments. This is because the latter has the same corrective role as the bonus tax, but does not collect tax revenue (cf. Keen, 2011). Therefore, we confine our analysis of policy instruments in this paper to a bonus tax.

Finally, we assume that the governments of banks' home countries $i \in \{1, 2\}$ choose their tax policies non-cooperatively. This implies that they are subject to similar competitive forces, arising from the international mobility of bank managers, as the national banking sectors are. However, the incentive structures of banks and governments are different. While attracting additional managers is always privately profitable for banks, governments may lose from an increased size of their national banking sector when the expected bailout costs exceed bonus tax revenue. For analytical tractability, our benchmark model focuses on fully symmetric countries. The symmetry assumption allows us, in particular, to directly compare the policies chosen under international tax competition, and those that are Pareto optimal for the region as a whole.¹²

Our following four-stage analysis proceeds by backward induction. Section 3 studies the decisions of bank managers. In Section 3.1, we derive the managers' effort- and risk-taking choices in Stage 4. Section 3.2 analyzes the managers' migration decisions in Stage 3. Section 4 then turns to the banks' optimal remuneration schemes in Stage 2 of the game, differentiating between the two regimes introduced above. Finally, in Section 5 we derive the governments' non-cooperative choice of bonus taxes in Stage 1.

¹¹This government objective will be generalized in the extensions (Section 6.1) by incorporating bank profits into the government's objective.

 $^{^{12}}$ In Section 6.3, we numerically analyze one instance of asymmetric tax competition where capital requirements and thus implicit government subsidies differ between countries.

3 The decisions of bank managers

3.1 Effort- and risk-taking choices

In Stage 4, managers choose their effort and risk-taking levels. All managers respond in the same way to a given remuneration scheme. Taking effort and risk involves private, nonmonetary costs for the manager.¹³ For analytical tractability, we assume that these cost functions are quadratic and given by $c^e(e) = \eta e^2/2$ and $c^r(r) = \mu r^2/2$. Due to these private costs, managers will neither exert enough effort nor take enough risk from the point of view of bank owners. Effort and risk-taking decisions are not observable. However, bank owners can mitigate the principal agent problem by a bonus payment z_i in the high return state, which occurs with higher probability p^h when effort and risk-taking are increased.

Managers located in country *i* maximize their location-specific utility u_i , which is the excess of expected bonus and fixed wage payments over the private costs of effort and risk-taking. Using (1) gives

$$u_i = p_i^h z_i + w_i - c^e(e_i) - c^r(r_i) = (\alpha e_i + \beta r_i) z_i + w_i - \frac{\eta e_i^2}{2} - \frac{\mu r_i^2}{2}.$$
 (2)

Maximizing (2) with respect to the managers' choice variables e_i and r_i yields

$$e_i = \frac{\alpha z_i}{\eta}, \qquad r_i = \frac{\beta z_i}{\mu}.$$
 (3)

Hence the managers' effort level e_i depends positively on the bonus payment z_i , and negatively on the effort cost parameter η . Analogously, the risk level r_i chosen by managers in country *i* is increasing in the bonus payment z_i and it is falling in the risk cost parameter μ . The fixed wage w_i does not affect managers' optimal effort or risk-taking decisions.

Using (3) in (1), we can derive the equilibrium probabilities of the states $\{h, m, l\}$:

$$p_i^{h*} = \alpha e_i + \beta r_i = \left[\frac{\alpha^2}{\eta} + \frac{\beta^2}{\mu}\right] z_i \equiv \gamma z_i, \tag{4a}$$

$$p_i^{m*} = p_0^m - \frac{\beta}{\mu} z_i, \tag{4b}$$

¹³As in Besley and Ghatak (2013), private costs of risk taking can be seen as the (psychological) costs of seeking out risk-taking opportunities above a 'natural' or benchmark level. This natural risk is here normalized to zero, but it could equally be set at a positive level without affecting any results.

$$p_i^{l*} = p_0^l + \left[\frac{\beta}{\mu} - \gamma\right] z_i \equiv p_0^l + \delta z_i.$$
(4c)

In eq. (4a), we have introduced the parameter $\gamma > 0$ to summarize the marginal effect of the bonus payment on the probability of a high return. This consists of two effects. A higher bonus leads to more effort and to more risk-taking, which both increase p^h . Similarly, in eq. (4c) the parameter δ summarizes the marginal effect of the bonus on the low return probability p^l . The sign of δ is ambiguous, in general. On the one hand, a higher bonus leads to more risk-taking, which increases p^l . On the other hand, a higher bonus payment induces more effort and this reduces p^l . In our following analysis we assume that $\delta > 0$, implying that the effect of the bonus on managers' risk-taking dominates the effect on managerial effort. Thus a higher bonus increases the probability of a low return.¹⁴ Finally, the effect of the bonus on the medium return in (4b) is unambiguously negative.

Finally, substituting (3) in (2) gives the location-specific utilities of managers:

$$u_{i}^{*} = \left[\frac{\alpha^{2}}{\eta} + \frac{\beta^{2}}{\mu}\right] \frac{z_{i}^{2}}{2} + w_{i} \equiv \frac{\gamma z_{i}^{2}}{2} + w_{i}.$$
(5)

Both a higher bonus and a higher fixed wage in country i increase the utility of managers working in this country.

3.2 The migration decision

In Stage 3 managers take the bonuses z_i and fixed wages w_i as given and choose whether to work in country 1 or in country 2. Managers maximize their gross utility, which consists of the location-specific utility in (5), and the non-monetary attachment to a particular country. There are a total of $2\bar{N}$ managers in the region, which are all employed in one of the two countries in equilibrium:

$$N_1 + N_2 = 2\bar{N}.$$
 (6)

Managers differ only in their country preferences. More precisely, managers are of type k, where k is the relative attachment to country 1 and we assume that k is distributed uniformly along $[-\bar{N}, +\bar{N}]$. Other things equal, all managers with k > 0 prefer to work in country 1, whereas managers with k < 0 prefer to work in country 2.

¹⁴This assumption corresponds to empirical evidence that higher risk-taking incentives are positively correlated with the probability of losses. See e.g. Cheng et al. (2015).

A common interpretation is that country 1 is the home country for all managers with k > 0, whereas country 2 is the home country for all managers with k < 0. We scale the location preference parameter k by the constant a. This constant captures the cultural, institutional and geographical distances between the two countries, where a large parameter a stands for large cross-country differences.¹⁵ The gross utility U_i of a manager of type k in country i is then

$$U_1(z_1,k) = u_1^*(z_1) + ak, \qquad U_2(z_2) = u_2^*(z_2).$$
 (7)

All managers choose to work in the country that gives them the higher gross utility. We characterize the manager that is just indifferent between working in country 1 or in country 2 by the location preference \hat{k} . Equating U_1 and U_2 in (7) and using (5), we derive \hat{k} as a function of differences in bonus payments and fixed wages between the two countries:

$$\hat{k} = \frac{1}{a} \left[\frac{\gamma}{2} (z_2^2 - z_1^2) + (w_2 - w_1) \right].$$
(8)

Managers with $k \in [\hat{k}, \bar{N}]$ work in country 1 and managers with $k \in [-\bar{N}, \hat{k}]$ work in country 2. Given the uniform distribution of k, there will be $\bar{N} - \hat{k}$ managers in country 1 and $\bar{N} + \hat{k}$ managers in country 2. Using (8) then determines the number of managers in country i as a function of the differences in bonus payments and fixed wages:

$$N_i = \bar{N} + \frac{1}{a} \left[\frac{\gamma}{2} (z_i^2 - z_j^2) + (w_i - w_j) \right] \qquad \forall i, j \in \{1, 2\}, \ i \neq j.$$
(9)

The larger is the bonus of country i, relative to that of country j, the more managers will work in country i in equilibrium. The same holds for the fixed wage. To quantify the managers' response, we introduce the semi-elasticity of migration with respect to the fixed wage:

$$\varepsilon \equiv \frac{\partial N_i}{\partial w_i} \frac{1}{N_i} = \frac{1}{aN_i} \tag{10}$$

The migration elasticity ε is the higher, the weaker is the managers' attachment to a particular country (the lower is the parameter a).¹⁶ Moreover, the migration elasticity falls in our model when the total number of managers in the country rises.

¹⁵See van Veen et al. (2014) for empirical evidence that a higher cultural, institutional and geographical distance between a manager's nationality and a company's country of origin makes it less likely that a manager of that nationality is employed by the company.

¹⁶See Kleven et al. (2014) for an empirical estimate of the migration elasticity of high income earners with respect to the after-tax wage factor. Their results for foreigners working in Denmark suggest a migration elasticity between 1.5 and 2 for this group.

4 Banks' compensation choices

In Stage 2, we turn to the remuneration decisions made by the owners of the single bank in each country. The bank in country *i* sets the bonus z_i and the fixed wage w_i to maximize its expected after-tax profits (which accrue to its shareholders).¹⁷ The expected after-tax profit of the bank in country *i* is

$$\Pi_i = N_i \pi_i^D,\tag{11}$$

where the number of divisions, which equals the number of managers, is given in (9).

The expected profit of each division, π_i^D , is determined by the division's exogenous financing structure, its endogenous investment decision, and the endogenous work contract. Each division finances its unit investment by a combination of savings deposits and equity, where the shares of deposits and equity in total liabilities are s and (1-s), respectively. These shares are determined by minimum capital requirements in each country, which are exogenous to our analysis. Since government guarantees are confined to external funds (i.e., savings deposits), banks will always exhaust the permissible level of external funds. Therefore, the share s of deposit finance is directly fixed by each country's capital requirement. Insured depositors face no risk and receive a risk-free interest rate fixed at d.

With this specification, and recalling that the division's gross return is zero in the low state, the expected profit of each division is

$$\pi_i^D = p_i^{h*}[Y^h - sd - z_i(1+t_i)] + p_i^{m*}[Y^m - sd] - w_i - (1-s)d.$$
(12)

The first two terms in (12) give the division's profits for the high and the intermediate return, respectively. When the representative division realizes Y^h (with probability p_i^{h*}), it pays sd to its depositors. Moreover, in state h the bank pays the net bonus z_i to its manager and the proportional bonus tax $t_i z_i$ to country *i*'s government. In state m, the division receives a portfolio return of Y^m and pays back sd to its depositors. Bonuses are not paid in this state.

If a division obtains the low return $Y^{l} = 0$, then it is unable to pay back its depositors, and so is the entire bank, due to the perfect correlation between the divisional returns

¹⁷Our analysis thus abstracts from a wage bargaining process between the bank's owners and its top management. See e.g. Piketty et al. (2014) for an optimal tax analysis with wage bargaining in the closed economy.

(see Section 2). In this case the payments to depositors (sd) are covered by deposit insurance, and thus eventually by the taxpayers in country *i*. Hence, in state *l*, these payments do not enter the division's profit in (12).

However, the manager's fixed wage w_i is paid by the division in all states, implying that the bank's shareholders realize a loss, with a corresponding reduction in the value of their equity, if the low state occurs.¹⁸ We thus assume that taxpayer-financed deposit insurance schemes are available to the bank in the low state l, before the bank's equity is wiped out completely. Finally, the last term in (12) gives the opportunity costs of the bank's equity (1 - s), which is valued at the risk-free interest rate d for notational simplicity. Incorporating this last term implies that division profits represent excess profits beyond the normal return to equity in our analysis.

In the following we differentiate two regimes. In Section 4.1 we derive the bank's optimal bonus payment when bonuses are the only remuneration for bank mangers (Regime I). Section 4.2 turns to the case where both the bonus and the fixed wage are paid in the bank's optimum (Regime II). We defer to the beginning of Section 4.2. the discussion of how the equilibrium regime is determined by the model's exogenous parameters.

4.1 Regime I: Bonus payment only

We first consider the case where the fixed wage w_i is set to zero in both banks and bonus payments in the high state are the only remuneration for bank managers. This implies that the bonus has the simultaneous tasks to incentivize bank managers with respect to their risk and effort choices, and to attract managers from the foreign country. Setting $w_i = 0$ in (12), maximizing the bank's after-tax profits in (11) and (12) with respect to the bonus z_i , and using (4a)–(4b) gives

$$\frac{\partial \Pi_i}{\partial z_i} = \frac{\gamma z_i}{a} \,\pi_i^D + N_i \left[\gamma (Y^h - sd) - \frac{\beta}{\mu} (Y^m - sd) - 2\gamma z_i (1 + t_i) \right] = 0. \tag{13}$$

The first effect in eq. (13) is unambiguously positive. A higher bonus z_i enables the bank to attract more managers and thereby run more divisions. This increases bank profits for any given expected profit per division. In an interior optimum, the second effect in (13) must therefore be negative, implying that the (after-tax) profit per division falls when the bonus is increased. This occurs by increasing the bonus until its effect on the

¹⁸We assume that fixed wage payments are less than the value of the bank's equity, $w_i < (1-s)d$.

expected gross return of the division (the first two terms in the squared bracket) is less than the bank's gross-of-tax cost of the bonus (the last term in the squared bracket). Deriving the optimal bonus z_i from (13) gives

$$z_i^*|_{RI} = \frac{N_i \Omega}{2N_i (1+t_i) - (\pi_i^D/a)}, \qquad \Omega \equiv (Y^h - sd) - \frac{\beta}{\mu\gamma} (Y^m - sd) > 0, \qquad (14)$$

where the term Ω summarizes the marginal effects of the bonus (via the managers' effort and risk-taking choices) on the bank's expected gross return. Clearly, this term must be positive for the bank to choose a positive bonus in equilibrium. Note, moreover, that the increase in the bank's expected gross profit exceeds the social return to the induced changes in managerial behavior, because the social cost of a higher failure probability induced by the bonus are not incorporated in Ω .¹⁹ Since the bonus is rising in Ω , bonus incentives set by the bank will therefore be 'excessive' from a social welfare perspective, as a result of the government's guarantees.²⁰

If bonuses are the only source of managerial pay, they are also set higher than they would be in a closed economy. This results from the international mobility of bank managers. The effect of bonuses on the number of managers is strictly positive in our international setting [the positive first term in (13)], while it would be zero in autarky. The international competition for managers thus increases the marginal benefit of bonuses, other things being equal, and therefore drives up bonus payments in equilibrium. As is shown in (14) this effect is the stronger, the lower is the strength of managers' country preferences, as measured by the parameter a.

4.2 Regime II: Bonus payment and fixed wage

We now turn to the case where the remuneration of bank managers is composed of both success-related bonus payments and a positive fixed wage w_i .²¹

¹⁹If these costs were incorporated, the term $\delta sd/\gamma$ would have to be subtracted from Ω in (14).

 $^{^{20}}$ See Laeven and Levine (2009) for empirical evidence that government guarantees provide banks with an incentive to increase risk-taking and Adams (2012) for evidence that banks with a higher performance pay for CEOs were more likely to receive government support in the financial crisis.

 $^{^{21}}$ In the European Union, for example, the employees of credit institutions and investment firms identified as having a material impact on the institution's risk profile had a ratio of variable over fixed remuneration of 65% in 2014. The highest ratios of variable over fixed pay for this group of bank employees are found in asset management (100%) and in investment banking (89%). See European Banking Authority (2016).

The first-order condition for the optimal bonus in (13) remains unchanged in this regime. Maximizing bank profits in (11) and (12) with respect to the fixed wage gives

$$\frac{\partial \Pi_i}{\partial w_i} = \frac{\pi_i^D}{a} - N_i = \frac{1}{a} \left[\pi_i^D - \frac{1}{\varepsilon} \right] \le 0 \quad \forall i,$$
(15)

where the second step has used the migration elasticity in (10). Eq. (15) holds with equality, and the fixed wage is used as a component of managerial compensation, when the division profit π_i^D (i.e., the gain from attracting an additional manager) is high, or when the migration elasticity of managers is high $(1/\varepsilon \text{ is low})$. In this case, a small positive wage attracts a large number of additional managers to the bank, relative to the number of division managers to whom the higher wage must be paid. If a small fixed wage has a positive effect on firm profits, then the wage rate will be further increased, lowering division profits in (12) until the first-order condition (15) is met with equality. Therefore, if ε is sufficiently high to ensure that $w_i > 0$, this implies $\pi_i^D/a = N_i$ from the complementary slackness condition (15). Substituting this into (14) the optimal bonus in this regime simplifies to

$$z_i^*|_{RII} = \frac{\Omega_i}{1+2t_i} \quad \forall \ i. \tag{16}$$

Comparing the optimal bonus in (16) with that in the regime without a fixed wage [eq. (14)] shows that the negative term $-\pi_i^D/a$ is now missing from the denominator. Hence the optimal bonus payment will be reduced when the fixed wage can also be optimally chosen by the bank. This reflects that the bonus is used only to affect managers' effort and risk-taking choices in this regime, whereas the fixed wage is the preferred instrument to attract additional managers from abroad.

In Appendix 1, we use equations (15) and (16) for both countries to derive the optimal fixed wage. This is given by

$$w_i^*|_{RII} = p_0^m (Y^m - sd) - (1 - s)d + \frac{1}{6}\gamma \Omega^2 \left[\frac{4t_i - 1}{(1 + 2t_i)^2} + \frac{1}{(1 + 2t_j)}\right] - \bar{N}a.$$
(17)

The last term in (17) shows that an increase in manager mobility (i.e., a decrease in a) increases the fixed wage. Since the task of attracting managers from abroad is shifted to the fixed wage instrument, a low value of a, and hence a weaker attachment to home, will now be reflected in a higher fixed wage that banks in each country offer to attract mobile managers.

In both regimes, increasing international mobility will thus lead to a higher overall compensation of managers in equilibrium. We summarize our results in: **Proposition 1** For given bonus taxes t_i , the more mobile managers are between countries (the lower is a), the higher is the overall remuneration of managers in equilibrium. In Regime I, the higher pay is reflected in a higher bonus payment, whereas in Regime II the premium is paid in the form of a higher fixed wage.

From Proposition 1, the higher international mobility of managers in recent decades (see Greve et al. 2015) can thus provide a possible explanation for the concurrent increase in banker compensation. This explanation complements the one given in the existing literature, which has focused on increased competition in national banking sectors as an explanation for the rise in managerial pay (Thanassoulis, 2012; Bannier et al., 2013). The further effects of higher international manager mobility depend, however, on which part of managerial compensation is raised. When international competition for managers occurs primarily via bonus payments (Regime I), the resulting higher bonuses will increase the risk-taking decisions of managers [see eq. (3)], and this will raise the probability of bank failures. In this case there is thus a direct, negative impact of the higher international mobility of bank managers on the stability of the financial system. In contrast, if the higher compensation occurs primarily through an increased fixed wage, then the higher international mobility of managers has no repercussions on their risk-taking choices in equilibrium.

4.3 The effects of bonus taxes on managerial remuneration

Before turning to governments' non-cooperative tax choices, we derive the effects that bonus taxes in each of the two countries have on managerial remuneration, and on the equilibrium number of managers working in each country. These effects will differ for the two regimes introduced above.

Regime I $(w_i = 0)$: In Regime I, an increase in the bonus tax changes bonus payments in both countries. Using (13) for both countries *i* and *j* and solving the system of two equations, Appendix 2 derives the effects of changes in bonus tax rates t_i and t_j on bank *i*'s optimal bonus payment z_i . This leads to:

$$\left. \frac{\partial z_i}{\partial t_i} \right|_{RI} < 0, \qquad \left. \frac{\partial z_i}{\partial t_j} \right|_{RI} < 0.$$
 (18)

The direct, first effect in (18) is straightforward. A higher bonus tax in country i makes bonus compensation more expensive for the bank in country i and thus reduces the optimal bonus payment z_i .²² Interestingly, however, a higher foreign tax t_j will also reduce the optimal bonus in country *i*. This is because a rise in t_j reduces the bonus payment z_j paid in country *j*. This reduces the attractiveness of working in country *j* and drives more managers to work in country *i*. Thus, an increase in z_i now has a negative effect on profits in more divisions in country *i*'s bank, whereas the marginal effect of z_i on attracting additional managers to country *i* is independent of the number of managers in country *i* [see eq. (9)]. On net, therefore, bonuses become more costly for banks in country *i* when country *j* raises its bonus tax.

Since the bonus is the only form of managerial pay in this regime, tax-induced changes in bonus payments directly affect the equilibrium number of managers in each country. From (9), this effect is given by

$$\frac{\partial N_i}{\partial t_i}\Big|_{RI} = \frac{\gamma}{a} \left(z_i \left. \frac{\partial z_i}{\partial t_i} \right|_{RI} - z_j \left. \frac{\partial z_j}{\partial t_i} \right|_{RI} \right) < 0.$$
(19)

This must be negative in a symmetric equilibrium with $z_i = z_j$, because $\partial z_i / \partial t_i$ and $\partial z_i / \partial t_j$ are both negative from (18), but $|\partial z_i / \partial t_i| > |\partial z_j / \partial t_i|$ follows from the stability of the equilibrium. Hence the net effect in (19) describes the equilibrium decrease in the number of managers working in country *i* when this country increases its tax rate and the bonus paid by country *i*'s bank falls by more than the bonus paid in country *j*.

Regime II $(w_i > 0)$: In Regime II, a bonus tax in country *i* only affects the bonus payment in this country, but it also changes the fixed wage in both banks. For the bonus payment, we get from (16):

$$\frac{\partial z_i}{\partial t_i}\Big|_{RII} = \frac{-2\Omega_i}{(1+2t_i)^2} < 0, \qquad \frac{\partial z_i}{\partial t_j}\Big|_{RII} = 0.$$
(20)

As in Regime I, a higher bonus tax in country i, t_i , reduces the optimal bonus z_i paid by the bank in i. Country j's bonus tax t_j has no impact on the optimal bonus payment in bank i. This is because bonuses are only used to induce bankers' effort and risk-taking choices in this regime, whereas the competition for internationally mobile bank managers occurs via the fixed wage.

²²Tax incidence results might differ, if bank managers were strongly risk averse. Dietl et al. (2013) show that a bonus tax can increase the bonus payment in this case, because the tax has an insurance effect that counteracts the higher costs of the bonus. However, the empirical evidence on the risk preferences of bank managers suggests that they are risk-neutral, or only very mildly risk-averse (see Thanassoulis, 2012).

The effects of bonus taxes in both countries on bank *i*'s fixed wage w_i are derived from (17):

$$\frac{\partial w_i}{\partial t_i} = \frac{4\gamma \Omega_i^2}{3} \left[\frac{(1-t_i)}{(1+2t_i)^3} \right], \qquad \frac{\partial w_i}{\partial t_j} = \frac{-\gamma \Omega^2}{3(1+2t_j)^2} < 0.$$
(21)

As long as $t_i < 1$, a higher bonus tax in country *i* will increase the fixed wage paid by bank *i*. Effectively, bank *i* shifts the compensation of its managers away from the more expensive bonus payment [see (20)] and towards the fixed wage, which is not covered by the additional tax.²³ When $t_i > 1$, however, the effect is turned around and a higher bonus tax reduces the fixed wage. Intuitively, in this case the high bonus tax makes it so costly to incentivize managers that it becomes less attractive for the banks to attract managers from abroad. Hence banks reduce their fixed wage. In contrast, an increase in country *j*'s bonus tax will always decrease bank *i*'s fixed wage. The higher bonus tax in *j* drives managers to country *i*. This increases the marginal cost of the wage w_i , which has to be paid to more managers, but it does not change the number of additional managers that a marginal increase in w_i can attract. The argument here is thus analogous to the one explaining that $\partial z_i/\partial t_j < 0$ in Regime I [eq. (18)].

These results can be used to derive the change in N_i caused by the bonus tax. Differentiating (9) with respect to t_i leads to

$$\frac{\partial N_i}{\partial t_i}\Big|_{RII} = \frac{1}{a} \left[\gamma z_i \frac{\partial z_i}{\partial t_i} \Big|_{RII} + \frac{\partial w_i}{\partial t_i} - \frac{\partial w_j}{\partial t_i} \right] = \frac{-\gamma \Omega^2}{3a(1+2t_i)^2} < 0,$$
(22)

where the second step follows from substituting in (20) and (21). In sum, the negative effect of a bonus tax in i on the bonus compensation of country i's managers dominates the changes in fixed wages in both countries. Therefore, a higher bonus tax in country ialso causes an outflow of managers in this regime.

5 International competition in bonus taxes

5.1 Governments' bonus tax decision

In Stage 1, governments set the bonus tax t_i that maximizes their net tax revenue W_i . In our model, net tax revenues are given by the expected bonus tax revenues minus

 $^{^{23}}$ Our benchmark analysis abstracts from general income taxes that would fall, at a uniform rate, on all forms of managerial compensation. In Section 6.1 we discuss how our results are affected when an additional income tax on managers' overall compensation is introduced.

expected bailout costs. Expected bonus tax revenue is collected from N_i managers in the domestic bank, multiplied by the expected bonus tax revenue per manager $p_i^h t_i z_i$. Bailout costs arise from compensating the depositors of the domestic bank in the event that the bank fails. They are obtained by multiplying the number of divisions N_i of the domestic bank with the expected bailout costs per division $p_i^l sd$.

We introduce F_i as the *net fiscal value* of a manager in country *i*, which equals expected tax income minus expected bailout costs per division. The government's net tax revenue is then given by

$$W_i = N_i F_i, \qquad F_i \equiv \left[p_i^{h*} t_i z_i - p_i^{l*} sd \right].$$
(23)

Importantly for our analysis, the fiscal value of a manager can be positive or negative. It is positive if, in the government's tax optimum, the revenue from taxing the manager's bonus exceeds the expected bailout costs for the government when the manager's division fails. This is more likely when the the share of subsidized deposit financing sis low (i.e., the bank's equity requirement is high), and when business conditions in the banking sector are favorable (the 'baseline' failure probability p_0^l in eq. (1) is low). It is also possible, however, that the reverse set of condition holds and s and p_0^l are both large. In this case the fiscal value of a manager, and hence also the total net domestic tax revenue W_i , is negative, even when the government chooses its bonus tax optimally. In the case where $W_i < 0$, we assume that there are unmodelled and fixed benefits for the economy from having a domestic banking sector, even though the exact size of the domestic bank does not matter for the real economy.²⁴ The non-tax benefits of having a domestic bank will then cause each government to accept negative net tax revenues from the banking sector, if fiscal conditions are unfavorable. Therefore, each government will still solve an interior tax optimization problem, rather than shutting down its domestic banking sector entirely.

Maximizing net tax revenue as given in (23) with respect to t_i then gives

$$\frac{\partial W_i}{\partial t_i} = N_i \frac{\partial F_i}{\partial t_i} + F_i \frac{\partial N_i}{\partial t_i} = N_i \left[z_i^2 \gamma + 2\gamma z_i t_i \frac{\partial z_i}{\partial t_i} - \delta sd \frac{\partial z_i}{\partial t_i} \right] + F_i \frac{\partial N_i}{\partial t_i} = 0.$$
(24)

We assume in the following that an interior tax optimum exists for each government. For the simpler Regime II, it is straightforward to derive the conditions under which

²⁴One possible setting that is in line with this assumption is that the production sector in each country can obtain credit from either the domestic or the foreign bank. In the complete absence of a domestic bank, however, the access to credit is either limited for the domestic economy, or it becomes discretely more expensive as a result of the foreign bank's monopoly power.

the government's objective W_i is a strictly concave function of the bonus tax rate t_i , and an interior tax optimum must exist. These conditions are derived in Appendix 3 and are summarized in:

Proposition 2 In Regime II, an interior tax optimum with $0 < t^* < \infty$ exists for both governments, and for both positive and negative fiscal values of bank managers $F_i \leq 0$, if the following condition holds:

$$\gamma \Omega \left(1 - \frac{\varepsilon p_0^l s d}{3} \right) - 2\delta s d > 0.$$

Proof: See Appendix 3.

Intuitively, the government's bonus tax revenue is unambiguously concave in the bonus tax rate. Therefore, a first condition for W_i to be strictly concave in t_i is that the effect of the tax rate change on bonus tax revenue dominates the effect on the government's bailout costs. This will be the case if the marginal effect of the bonus on the probability of the high return state (as parameterized by $\gamma > 0$) dominates the marginal effect of the bonus on the probability of the low state (as parameterized by $\delta > 0$).²⁵ The second condition ensuring an interior solution is that the tax-induced net revenue changes resulting from the changed incentives of existing managers in a country dominate the revenue changes resulting from the inflow or outflow of managers. This condition will be met when the migration elasticity of managers, ε , is not too large.²⁶

If the conditions summarized in Proposition 2 are met, each country will choose an interior tax rate t^* in its tax optimum, for both positive and negative fiscal values of managers F_i . From the first-order condition (24) we can then interpret the fundamental trade-off for governments in more detail. The derivatives $\partial z_i/\partial t_i$ and $\partial N_i/\partial t_i$ are regime-specific, but we have shown in Section 4.3 that both derivatives have the same sign in Regimes I and II. Starting with the second term in (24), a bonus tax causes an outmigration of managers in both regimes [from eqs. (19) and (22)]. Therefore the second term in (24) has the opposite sign as the fiscal value of a manager, F_i .

²⁵Recall from our discussion in Section 3.1 [eqs. (4a)–(4c)] that γ is unambiguously positive from both the effort and risk-taking choices of bank managers. In contrast, the effects of effort and risk taking on the parameter δ are mutually offsetting.

²⁶Note that moderate migration elasticities required for an interior tax equilibrium do not preclude an equilibrium where fixed wages are part of managerial compensation [see eq. (15)]. In particular, when the division profits π_i^D are sufficiently high due to a high exogenous return Y^h , then even moderate migration elasticities will ensure that (15) holds with equality, implying a Regime II equilibrium.

The sign of the first term depends on the sum of three terms in the squared bracket, which give the change in net tax revenue for a representative division $(\partial F_i/\partial t_i)$. The first term gives the direct effect of a tax increase at an unchanged tax base and is clearly positive. The second term is unambiguously negative, because a higher bonus tax reduces bonus payments, and hence the bonus tax base, in both regimes [eqs. (18) and (20)]. The third term is again positive, as the tax induced fall in bonus payments lowers the probability of bank default and thus reduces taxpayer losses when $\delta > 0$.

We can thus differentiate two cases. If the fiscal value of a manager, F_i , is positive, then the outmigration of managers caused by the tax increase leads to a negative second effect in (24). In an interior optimum, the first term in (24) must therefore be positive, on net. This requires that the negative second term in the squared bracket is small, implying a low bonus tax rate t_i in equilibrium.

In the opposite case where $F_i < 0$, the falling number of managers and bank divisions creates a net revenue gain for the government (i.e., a reduction in the net subsidies paid to the banking sector) from the second effect in (24). In this case the increase in the bonus tax rate must therefore reduce the net revenue obtained from each division in an interior tax optimum, and the first term in (24) must be negative. This requires that the negative second term in the squared bracket is large, implying a high bonus tax rate t_i in the non-cooperative tax equilibrium.

5.2 Are non-cooperative bonus taxes too high or too low?

The next step is to compare the optimal bonus taxes chosen by non-cooperating governments to the bonus tax rates that would be optimal from a regional welfare perspective. We start from an interior, symmetric equilibrium where $\partial W_i/\partial t_i = 0$ holds for both countries $i \in \{1, 2\}$. Since countries are symmetric, we can define regional welfare as the sum of national welfare levels

$$W_W = W_i + W_j \quad \forall \ i, j \in \{1, 2\}, i \neq j,$$
(25)

where W_i is given in eq. (23). Choosing t_i so as to maximize regional welfare, eq. (25) would imply $\partial W_W / \partial t_i = 0$. The nationally optimal bonus taxes derived in the previous section are instead chosen so that $\partial W_i / \partial t_i = 0$ [see eq. (24)]. Hence, any divergence between nationally and globally optimal bonus taxes is shown by the effect of country *i*'s policy variable t_i on the welfare of country *j*. If $\partial W_j / \partial t_i > 0$, bonus taxes chosen at the national level are 'too low' from a global welfare perspective, as an increase in t_i would generate a positive externality on the welfare of country j. The reverse holds if $\partial W_j/\partial t_i < 0$. In this case the externality on the foreign country is negative and nationally chosen bonus taxes are 'too high' from a regional welfare perspective.

Using this argument and employing symmetry, which implies $\partial W_j/\partial t_i = \partial W_i/\partial t_j$, we differentiate (23) with respect to the foreign tax rate t_j . This gives:

$$\frac{\partial W_i}{\partial t_j} = (2t_i\gamma z_i - \delta sd) \frac{\partial z_i}{\partial t_j} N_i + F_i \frac{\partial N_i}{\partial t_j}.$$
(26)

We analyze eq. (26) for the two different regimes in our model.

Regime I $(w_i = 0)$: In Regime I, there are two main externalities of bonus taxes. The first term in eq. (26) stems from the fall in country *i*'s bonus payment z_i that is induced by the tax increase in country *j* [see eq. (18)]. The falling bonus in *i* is associated with lower bonus tax revenues, but also with lower expected bailout cost. Hence this effect is ambiguous, in general. The second effect in (26) is driven by the migration decision of managers. Since the sum of managers working in one of the two countries is fixed, we get $\partial N_i/\partial t_j = -\partial N_j/\partial t_j > 0$ from (19). A rise in t_j decreases the bonus in country *j* by more than in country *i*, and thus increases the number of managers in country *i* in equilibrium.

The overall sign of the second term thus hinges critically on the sign of F_i , the *fiscal* value of a manager. If this term is positive, a bonus tax increase in country j will benefit country i through the immigration of managers, who are net contributors to tax revenues. In this case the net fiscal externality is likely to be positive, implying that bonus taxes set in the non-cooperative equilibrium are lower than the bonus taxes that would be chosen under policy coordination. This is the conventional case of a 'race to the bottom' in the setting of bonus taxes. From (23) this scenario will be the more likely, the lower are the implicit government subsidies through deposit insurance (the lower is s), and the lower is the baseline probability of a low return, p_0^l .

Conversely, if the expected bailout costs for governments dominate the expected revenue from a bonus tax so that $F_i < 0$, then the second effect in (26) is negative. In Appendix 4 we show that $F_i < 0$ is indeed a *sufficient* condition for the overall externalities of bonus taxation to be negative. Non-cooperatively set bonus taxes are thus unambiguously higher than those in the coordinated equilibrium and there is a 'race to the top' in bonus taxation. Intuitively, managers are unwanted by governments in this case, as the expected bailout costs for the government dominate the revenue potential from bonus taxation. Hence each country attempts to drive bank managers to the other country by means of a high bonus tax, thus shifting fiscal risks from domestic to foreign taxpayers. This case is more likely, if the probability of government guarantees for the banking sector are large (s is high), and if business conditions for the banking sector are unfavorable (p_0^l is high).

Regime II $(w_i > 0)$: In Regime II the first term on the right-hand side of (26) equals zero, as country j's bonus tax has no impact on the bonus paid by country i's bank [see eq. (20)]. In the second term of (26), the effect of a change in t_j on the number of managers in country i is again positive [see eq. (22)]. Hence in Regime II the sign of the fiscal externality is always equal to the sign of the fiscal value per manager, F_i . If this term is positive, a higher bonus tax in country j will benefit country i through the immigration of managers and a 'race to the bottom' occurs. If F_i is negative, the immigration of managers associated with a rise in country j's bonus tax is instead harmful for country i and a 'race to the top' results.

We can thus conclude that the effects of tax competition are very similar in the two regimes. We summarize our results in this section in:

Proposition 3 When non-coordinated bonus taxation leads to a symmetric, interior tax equilibrium, the following holds:

(i) In both regimes, a positive fiscal value of a manager $(F_i > 0)$ is a necessary condition for non-coordinated bonus taxes to be below their globally optimal levels, and hence for a 'race to the bottom' to occur. In Regime II $(w_i > 0)$ the condition is also sufficient.

(ii) In both regimes, a negative fiscal value of a manager $(F_i < 0)$ is a necessary and sufficient condition for non-coordinated bonus taxes to be above their globally optimal levels, and hence for a 'race to the top' to occur.

Proof: See Appendix 4.

5.3 Discussion

Our results in Proposition 3 incorporate two different settings. When the fiscal value of managers is positive, governments undertax bonuses, relative to the globally efficient level, in an attempt to attract more 'fiscally valuable' bank managers. In this case,

therefore, governments will not fully correct the distortion arising from the excessive bonus incentives set by banks, as a result of their limited liability. When the fiscal value of bank managers is negative, however, government's incentives to correct for the banks' limited liability and their strategic incentive to reduce bank size are mutually reinforcing. Hence, each country overtaxes bonuses in this case, in order to drive 'fiscally harmful' managers to the other country.

The case where bonus taxes are below their efficient levels corresponds to the setting that is well known from the tax competition literature (see Keen and Konrad, 2013, for a recent survey). The recent literature has generalized this result to non-linear income tax competition between governments pursuing a redistributive objective (Lehmann et al., 2014; Lipatov and Weichenrieder, 2015), and it has shown that tax competition may even eliminate all taxes on the high-skilled when they are perfectly mobile across countries (Bierbrauer et al., 2013). More generally, Sinn's (1997) conclusion applies in this setting that governments in competition will be unable to fully correct the externalities that arise from (allocative or distributional) market failures.

The opposite setting with inefficiently high bonus taxes rarely occurs in the tax competition literature, which typically excludes risk-taking decisions, and hence the possibility of negative returns. This setting has some similarities with the NIMBY (Not In My Backyard) scenario, however, that is known from the taxation of environmentally hazardous plants or products (e.g. Markusen et al., 1995). The main difference to this scenario is that the negative externalities in our case are fiscal ones: High bonus taxes are used by each country to shift the fiscal risks associated with bailout guarantees from domestic to foreign taxpayers.

The ambiguity about the direction of tax competition is particularly relevant in the banking sector, due to the possibility that bank managers cause fiscal losses for taxpayers. In our model, governments use price signals (i.e., taxes) to change the behavior of banks and, via the change in bonus payments, the behavior of managers in the direction of lower risk-taking. Similar effects can also be obtained by forcing banks to hold more equity capital by means of minimum capital requirements. A small literature has studied regulatory competition in capital standards and has typically found that this competition leads to a 'race to the bottom' in capital standards when governments focus primarily on maximizing domestic bank profits (Acharya, 2003; Sinn, 2003; Dell'Ariccia and Marquez, 2006). Recently, Haufler and Maier (2016) have shown, however, that regulatory competition in capital standards will instead lead to a 'race to the top' when the governments' objective function is broadened and also includes fiscal risks as well as consumer surplus, which is affected by the overall availability of credit. In sum, therefore, the direction of regulatory competition is ambiguous in this literature, similar to our results for bonus taxation.

6 Extensions

In the following we extend our model in various directions. In Section 6.1 we incorporate bank profits into the welfare functions of governments. Section 6.2 then investigates the situation in a banking union, where the two symmetric countries internalize a share of each other's bailout costs. Our focus in both sections lies on the fiscal externalities that are added to our benchmark analysis, and how these affect the direction of bonus tax competition. Finally, Section 6.3 studies, by means of numerical analyses, the asymmetric tax competition equilibrium that arises when exogenous capital requirements, and hence the shares of subsidized savings deposits, differ between countries.

6.1 Bank profits in the government's objective

So far we have included only bonus tax revenue and bailout costs in the welfare functions of governments. We now analyze the case where each government additionally takes into account a share θ of the domestic bank profits Π_i . The objective function thus changes to:

$$\tilde{W}_i = N_i F_i + \theta N_i \pi_i^D. \tag{27}$$

The new second term on the right-hand side gives the income that domestic capital owners derive from the profits of the domestic banking sector. The welfare weight θ thus jointly reflects the share of the domestic banking sector that is owned by domestic residents, and the relative valuation of this profit income in the government's objective.

In **Regime I**, differentiating (27) with respect to t_j gives

$$\frac{\partial \tilde{W}_i}{\partial t_j}\Big|_{RI} = (2t_i\gamma z_i - \delta sd) \frac{\partial z_i}{\partial t_j} N_i + F_i \frac{\partial N_i}{\partial t_j} + \theta \left\{ \underbrace{\left[\underbrace{N_i \frac{\partial \pi_i^D}{\partial z_i} + \pi_i^D \frac{\partial N_i}{\partial z_i}}_{=0} \right]}_{=0} \frac{\partial z_i}{\partial t_j} + \pi_i^D \frac{\partial N_i}{\partial t_j} \right\}.$$
(28a)

The first two terms in (28a) correspond to those in our benchmark model [eq. (26)]. The new terms resulting from the change in country i's profit income are due to the

effects that a higher tax in country j has on bonus payments by both banks. The effects induced by the change in bank i's bonus payment z_i must sum to zero from this bank's optimal bonus choice [see eq. (13)], so that only the effects operating through bank j's bonus z_j remain (the second term in the curly bracket). These effects are unambiguously positive for country i as the higher tax in j induces a reduction in z_j . This in turn drives managers to country i [see eq. (19)] and increases the size of country i's bank.

In **Regime II**, differentiating (27) gives

$$\frac{\partial \tilde{W}_i}{\partial t_j}\bigg|_{RII} = F_i \frac{\partial N_i}{\partial t_j} + \theta \pi_i^D \frac{\partial N_i}{\partial t_j} .$$
(28b)

The first term in (28b) corresponds to the only fiscal externality in our benchmark model, as $\partial z_i/\partial t_j = 0$ holds in this regime. The additional term on bank profits in country *i* results again from the changed bank size in country *i* following the increase in t_j . This effect is positive from eq. (22). The simultaneous change in w_i caused by a rise in t_j has no first-order welfare effect on country *i*, because w_i is optimally chosen by country *i*'s bank [eq. (15)].

In sum, adding bank profits to the government's objective adds a positive term to the fiscal externalities in both regimes. This implies that a 'race to the bottom' becomes more likely under this extension. Intuitively, by increasing its tax rate, the government of country j causes some bank managers to move to country i. This will increase *bank profits* in country i, even when the net contribution of bank managers to country i's tax revenues is negative.

A related extension is to add a general income tax, levied at an exogenous rate, that bank managers have to pay on all their income (i.e., the fixed wage and the bonus). This extension is therefore relevant only in Regime II. Incorporating income taxes does not add an additional component to the government's objective, as in (28b), but it adds a positive revenue term to F_i . This makes it more likely that $F_i > 0$ holds in equilibrium, and hence it increases the probability that the single remaining externality in (28b) is positive. Therefore, adding exogenous income taxes for bank managers to the model also increases the likelihood of a 'race to the bottom' in bonus taxation.

6.2 Joint liability of bailout costs

Another relevant extension of our benchmark model is to incorporate joint liability of the two countries in the case of individual bank failures. In the Euro area, such a scheme exists as the so-called 'Single Resolution Mechanism' within the EU's banking union. In the following we assume that taxpayers in the two countries jointly come up for the losses caused by each national bank.²⁷ We take ρ to be the share that taxpayers in country *i* pay for the expected losses of bank failures in country *j*, whereas $(1 - \rho)$ is the share of losses that taxpayers in each country pay for the bank losses in their own country. Joint liability of bailout costs then implies

$$\hat{W}_i = N_i \left[T_i - (1 - \rho) B_i \right] - \rho N_j B_j \quad \forall i \neq j,$$
(29)

where T_i is the tax revenue per manager and B_i is the bailout cost per manager in *i*:

$$T_i \equiv p_i^{h*} t_i z_i, \qquad B_i \equiv p_i^{l*} sd \quad \forall \ i.$$

To analyze the fiscal externalities associated with bonus taxation in **Regime I**, we differentiate (29) with respect to t_j . This gives

$$\frac{\partial \hat{W}_i}{\partial t_j} = \frac{\partial N_i}{\partial t_j} \left[T_i - (1-\rho)B_i \right] + N_i \left[\frac{\partial T_i}{\partial t_j} - (1-\rho)\frac{\partial B_i}{\partial t_j} \right] - \rho \left[B_j \frac{\partial N_j}{\partial t_j} + N_j \frac{\partial B_j}{\partial t_j} \right].$$
(30)

To see how the fiscal externalities change with respect to the collectivization of bailout costs, we differentiate (30) with respect to ρ . This gives, after using symmetry and summarizing terms

$$\frac{\partial \hat{W}_i/\partial t_j}{\partial \rho} = 2B_i \frac{\partial N_i}{\partial t_j} - N_i \delta sd \left(\frac{\partial z_j}{\partial t_j} - \frac{\partial z_i}{\partial t_j}\right). \tag{31}$$

The first term in (31) is always positive, since $\partial N_i/\partial t_j > 0$. This effect captures that the negative externality from shifting bailout costs abroad via manager migration becomes smaller when the degree of collectivizing bailout costs is increased (ρ rises). The second term in (31) is also positive when $\delta > 0$. This is because an increase in t_j reduces the bonus in country j, thus reducing risk-taking and hence the expected losses arising in country j's banking sector. For a higher level of ρ a larger part of this net revenue gain is transferred to country i. Finally, from the stability condition $|\partial z_j/\partial t_j| > |\partial z_i/\partial t_j|$, the change in z_j induced by a rise in t_j dominates the counteracting effect from the simultaneous change in the bonus of bank i.

²⁷We thus abstract from insurance funds paid by the banking sector. In Europe, member states of the banking union are building up an EU-wide 'resolution fund', financed by levies on member states' banks. This fund, however, is built up only gradually and with a moderate overall target volume of 1% of the covered deposits of banks (55 billion Euro, based on the volume of deposits in 2010).

In **Regime II** the only difference is that the second term in (31) simplifies because $\partial z_i/\partial t_j = 0$ from (18). This, however, does not change any qualitative results. In sum, the effects in (31) are thus unambiguously positive in both regimes, implying that a higher degree of collectivizing bailout costs (a rise in ρ) increases the value of the net externality $\partial \hat{W}_i/\partial t_j$ and makes a 'race to the bottom' more likely. We summarize our results in the first two extensions of this section in:

Proposition 4 A 'race to the bottom' in bonus taxes becomes more likely, if

- (i) domestic bank profits receive a higher weight in the welfare function of governments $(\theta \text{ is increased}), \text{ or if}$
- (ii) bailout costs are more strongly collectivized between countries (ρ rises).

Proposition 4 (i) can be used to rationalize the development of bonus taxation in the recent past. In the aftermath of the 2008 financial crisis, large-scale bank bailouts occurred in many countries, implying that the net fiscal value per bank manager (F_i) was frequently negative. Given this experience, the protection of national taxpayers was the dominant concern in many countries, relative to the incentive to increase banking sector profits (i.e., θ was low). These conditions may explain why very high bonus tax rates were enacted, or at least prepared, in several OECD countries in 2009 and 2010 (see the introduction). In the following years, however, perceived risks for taxpayers fell and banking sector profits resumed, and with it the incentive to attract banking sector profits from abroad (a rise in θ). These developments may have caused the competition via bonus taxes to change directions, moving from a 'race to the top' to a 'race to the bottom', and to the repeal of previously enacted bonus taxes.

Moreover, from Proposition 4 (*ii*), collectivizing the costs of bank restructuring in the European banking union may further contribute to a 'race to the bottom' in bonus taxation when bonus taxes are set unilaterally and non-cooperatively. Setting a lower bound on bonus taxes or, alternatively, limiting bonus payments by regulatory means may thus be a desirable coordination measure complementing the banking union. And indeed, the latter occurred with the coordinated 2014 regulation limiting bonus payments in the EU to 100% of bankers' fixed salary.

6.3 Asymmetries between countries

Finally, we depart from the assumption that the two countries considered in our analysis are identical in all respects. More specifically, we consider cross-country differences in the share of deposit finance, s_i , and let country 1 have the higher share of deposit finance so that $s_1 > s_2$. This is most easily interpreted as a difference in the (exogenous) capital adequacy ratios for banks stipulated by national regulators, which are assumed to be less tight in country 1 as compared to country 2. Hence banks in country 1 receive higher implicit taxpayer subsidies, equal to $p_1^l s_1 d$ [cf. eq. (23)].

Our analysis in this subsection focuses on **Regime I**. Let us first consider the effects of a change in s_1 on the optimal bonus decision of the bank in country A. From the optimal bonus choice in (13) and using $\beta/\mu - \gamma \equiv \delta$ from (4a)–(4c), we get

$$\frac{\partial z_1}{\partial s_1} = \frac{\partial^2 \Pi_1 / \partial z_1 \partial s_1}{-(\partial^2 \Pi_1 / \partial z_1^2)} \quad \text{where} \quad \frac{\partial^2 \Pi_1}{\partial z_1 \partial s_1} = \frac{\gamma z_1}{a} (1 - p_1^l) d + N_1 d\delta. \tag{32}$$

Eq. (32) gives the change in the optimal bonus for fixed levels of bonus taxes t_i . It shows that the bank in country 1 will pay a higher bonus in response to the lower capital adequacy ratio (the increase in s_1). Firstly, when banks are financed by a larger share of savings deposits, the taxpayers of country 1 bear a larger share of the losses when the low state l is realized. This increased level of insurance, and hence profitability, will induce the bank in country 1 to offer a higher bonus in order to increase bank size. Secondly, the positive effect on the bonus is further increased when a higher bonus increases the likelihood of a low outcome ($\delta > 0$), as this is the event that is subsidized more heavily when s_1 rises. Hence $\partial z_1/\partial s_1 > 0$ follows unambiguously when $\delta \ge 0$. In this section we also allow for $\delta < 0$, however. In this case, the second effect changes its sign and the relationship between the deposit share s_1 and the bonus rate z_1 is no longer unambiguous, even for a fixed bonus tax rate t_1 .

To determine the effects of a rise in s_1 on the tax rate in country 1, we use the firstorder condition of country 1's tax rate in (24). The differentiation is carried out in Appendix 5. It is shown there that the overall effects depend critically on the sign of δ . When $\delta > 0$, so that a higher bonus primarily increases risk-taking, a rise in s_1 is likely to raise t_1 in the government's optimum. Intuitively, the increase in the insured deposit share s_1 increases the exposure of country 1's government to the losses occurring in state l. Therefore, the tax will be adjusted upward, in order to discourage bonus payments and minimize the likelihood of the low state. Conversely, if $\delta < 0$ a higher bonus primarily increases managerial effort and thus reduces the likelihood of the low state. In this case, it is therefore optimal for country 1's government to reduce the bonus tax, in order to boost bonus payments by the local bank.

Table 1 confirms these expectations with the help of some numerical analyses. We

α	δ	z_1	z_2	t_1	t_2
$s_1 = 0.8, s_2 = 0.5$					
0.40	0.180	0.326	0.348	1.326	1.068
0.45	0.095	0.450	0.462	0.884	0.778
0.50	0	0.589	0.593	0.615	0.581
0.55	-0.105	0.734	0.733	0.450	0.454
0.57	-0.150	0.791	0.788	0.403	0.418

Table 1: Bonus payments and tax rates for different shares of deposit finance

<u>Note:</u> Parameters held constant: $\beta = 0.5$, $p_0^l = 0.2$, $p_0^m = 0.8$, $\eta = 0.5$, $\mu = 0.5$, $Y^h = 3$, $Y^m = 1.5$, a = 0.1, d = 1.

specify a higher share of deposit finance in country 1 as compared to country 2 ($s_1 = 0.8$, $s_2 = 0.5$). In the examples we continuously increase the impact of managerial effort, α . Hence the parameter δ , which gives the net effect of changes in the bonus z_i on the probability of the low state, p_i^l [see eq. (4c)] switches from a positive to a negative sign. As long as $\delta \geq 0$, the bonus tax rate in country 1 exceeds that of country 2, as the government of country 1 has the higher fiscal exposure to the losses generated by its bank in the low state, and hence has the stronger incentive to reduce bonus pay by means of high taxes. In fact, the tax difference is so large that equilibrium bonus payments are lower in country 1, as compared to country 2, despite the positive direct effect of s_1 on z_1 given in (32). When δ turns negative, however, the tax difference turns around and country 1 levies the lower bonus tax. This is because bonus payments now primarily increase effort and thus reduce the probability of the low state, p^l . Again, it is country 1 which has the stronger incentive to minimize p^l , and hence it now charges a lower tax rate than country 2 in the asymmetric tax competition equilibrium.

7 Conclusion

In this paper we have incorporated international mobility of bank managers into a framework with two principal-agent problems. Banks choose their compensation structure, which consists of a fixed wage and bonus pay, so as to simultaneously induce managers to take the effort and risk choices desired by the principal, and to attract additional managers from abroad. In the event of failure, banks impose negative externalities on taxpayers, due to the existence of government guarantees. This gives rise to excessive bonus payments by banks, relative to those that would be socially optimal. Governments therefore choose bonus taxes to collect tax revenue and to counteract the distorted incentives of banks and their managers. In doing so governments, like banks, are subject to the international competition arising from bank managers' mobility.

In such a setting non-cooperative levels of bonus taxes can generally be above or below the levels that would be optimal under policy coordination. Therefore there can be a 'race to the bottom' or a 'race to the top' in bonus taxes. The 'race to the top' result arises if bank managers have a negative fiscal value for the jurisdiction in which they work, inflicting expected losses on taxpayers that exceed bonus tax revenues. In this case, governments set bonus taxes above their Pareto efficient levels, in order to shift fiscal risks from domestic to foreign taxpayers.

The 'race to the top' result is specific to highly paid agents that, despite being a source of tax revenue, take risky decisions without bearing the full cost of it. It may explain the wave of very high marginal taxes on bankers' bonuses in the immediate aftermath of the 2008 financial crisis. More recently, however, the perceived risks from bank failures have fallen again, while jobs and profits in the banking sector have gained new importance. This may have changed incentives for governments once more, in the direction of a 'race to the bottom'. In the newly created European banking union the costs of bank defaults are furthermore shared between its member states, strengthening the incentives to adjust bonus taxes downwards. The coordinated cap on bonus payments that EU countries have recently enacted can therefore be seen as a direct complement to the creation of the banking union.

Our analysis can be extended in various directions. A first possible extension is to extend the set of government instruments and endogenize regulatory policies, which have been taken as exogenous in the present analysis. To introduce a meaningful trade-off for governments with respect to minimum capital requirements, however, this extension would require to model a real sector that depends on bank loans to finance its investments. A further interesting extension is to consider bank managers that are overconfident (see Ho et al., 2016 for recent evidence) and therefore overvalue the bonus component of their compensation. What does this imply for the banks' optimal compensation structure, and for the tax incentives of governments? We plan to address these issues in future research.

Appendix

A.1 Derivation of equation (17)

The bank in country *i* chooses the bonus z_i and the fixed wage w_i , which depends on z_j and w_j . Hence the system of first-order conditions in (15) is interdependent, and given by

$$\frac{\partial \Pi_i}{\partial w_i} = \frac{1}{a} \left\{ p_i^{h*} [Y^h - sd - z_i(1+t_i)] + p_i^{m*}(Y^m - sd) - w_i - (1-s)d \right\}
- \bar{N} - \frac{1}{a} \left[\frac{\gamma}{2} (z_i^2 - z_j^2) + w_i - w_j \right] = 0,$$
(A.1)
$$\frac{\partial \Pi_j}{\partial w_j} = \frac{1}{a} \left\{ p_j^{h*} [Y^h - sd - z_j(1+t_j)] + p_j^{m*}(Y^m - sd) - w_j - (1-s)d \right\}
- \bar{N} - \frac{1}{a} \left[\frac{\gamma}{2} (z_j^2 - z_i^2) + w_j - w_i \right] = 0.$$
(A.2)

Solving both (A.1) and (A.2) for w_j , setting the two expressions equal to each other, and solving for w_i gives

$$w_{i} = \frac{1}{3}p_{j}^{h*}[Y^{h} - sd - z_{j}(1+t_{j})] + \frac{1}{3}p_{j}^{m*}(Y^{m} - sd) + \frac{2}{3}p_{i}^{h*}[Y^{h} - sd - z_{i}(1+t_{i})] + \frac{2}{3}p_{i}^{m*}(Y^{m} - sd) - (1-s)d - a\bar{N} + \frac{1}{6}\gamma(z_{j}^{2} - z_{i}^{2}).$$

Substituting in the equilibrium bonuses z_i and z_j from (16) and the equilibrium probabilities from (4a)–(4c) yields

$$\begin{split} w_{i} &= \frac{1}{3} \frac{\gamma \Omega_{j}}{(1+2t_{j})} \left[Y^{h} - sd - \frac{\Omega_{j}(1+t_{j})}{1+2t_{j}} \right] + \frac{1}{3} \left(p_{0}^{m} - \frac{\beta}{\mu} \frac{\Omega_{j}}{1+2t_{j}} \right) (Y^{m} - sd) \\ &+ \frac{2}{3} \frac{\Omega_{i}\gamma}{(1+2t_{i})} \left[Y^{h} - sd - \frac{\Omega_{i}(1+t_{i})}{1+2t_{i}} \right] + \frac{2}{3} \left(p_{0}^{m} - \frac{\beta}{\mu} \frac{\Omega_{i}}{1+2t_{i}} \right) (Y^{m} - sd) \\ &- (1-s)d - a\bar{N} + \frac{1}{6}\gamma \left[\left(\frac{\Omega_{j}}{1+2t_{j}} \right)^{2} - \left(\frac{\Omega_{i}}{1+2t_{i}} \right)^{2} \right]. \end{split}$$

Using $\Omega = Y^h - sd - \frac{\beta}{\mu\gamma}(Y^m - sd)$ from (14) and symmetry, this simplifies to

$$w_i = p_0^m (Y^m - sd) - (1 - s)d + \frac{\gamma \Omega^2}{6} \left[\frac{1}{(1 + 2t_j)} + \frac{4t_i - 1}{(1 + 2t_i)^2} \right] - \bar{N}a,$$
(A.3)

which corresponds to eq. (17) in the main text.

A2. Derivation of equation (18)

We start from the set of first-order conditions for the banks' optimal bonus payments in eq. (13):

$$\frac{\partial \Pi_i}{\partial z_i} \equiv H_i(z_i, z_j, t_i) = 0, \qquad \frac{\partial \Pi_j}{\partial z_j} \equiv H_j(z_i, z_j, 0) = 0, \quad i \neq j,$$

where z_i and z_j are the endogenous variables in Stage 2 and the tax rate t_i is an exogenous shifter.

Totally differentiating and employing matrix notation gives

$$\mathbf{A} \begin{bmatrix} dz_i \\ dz_j \end{bmatrix} = \begin{bmatrix} -(\frac{\partial H_i}{\partial t_i}) dt_i \\ 0 \end{bmatrix}, \qquad \mathbf{A} = \begin{bmatrix} \frac{\partial H_i}{\partial z_i} & \frac{\partial H_i}{\partial z_j} \\ \frac{\partial H_j}{\partial z_i} & \frac{\partial H_j}{\partial z_j} \end{bmatrix}$$

This can be solved for dz_i/dt_i and dz_j/dt_i to give

$$\frac{dz_i}{dt_i} = -\frac{1}{|A|} \frac{\partial H_j}{\partial z_j} \frac{\partial H_i}{\partial t_i} , \qquad \frac{dz_j}{dt_i} = \frac{1}{|A|} \frac{\partial H_j}{\partial z_i} \frac{\partial H_i}{\partial t_i} , \qquad |A| = \left(\frac{\partial H_j}{\partial z_j}\right)^2 - \left(\frac{\partial H_j}{\partial z_i}\right)^2 > 0$$
(A.4)

where |A| has used the symmetry of the model and |A| > 0 follows from the stability of the Nash equilibrium. The derivatives in (A.4) are obtained from the first-order condition (13), which is reproduced here for convenience:

$$\frac{\partial \Pi_i}{\partial z_i} \equiv H_i = \frac{\gamma z_i}{a} \,\pi_i^D + \left[N + \frac{\gamma}{2a}(z_i^2 - z_j^2)\right] \frac{\partial \pi_i^D}{\partial z_i} = 0 \quad \forall \ i \neq j. \tag{A.5}$$

The division profit π_i^D and its derivative with respect to z_i are given by

$$\pi_i^D = p_i^{h*} [Y^h - sd - z_i(1+t_i)] + p_i^{m*} [Y^m - sd] - (1-s)d,$$

$$\frac{\partial \pi_i^D}{\partial z_i} = \gamma [Y^h - sd - z_i(1+t_i)] - p_i^{h*}(1+t_i) - \frac{\beta}{\mu} (Y^m - sd) < 0, \qquad (A.6)$$

and (A.6) must be negative in an interior optimum from (A.5).

To sign the terms in (A.4) we assume first that the second-order condition for an optimal choice of z_i holds. This implies $\partial H_j/\partial z_j < 0$. From (A.5), we obtain by differentiation

$$\frac{\partial H_i}{\partial z_j} = \frac{-\gamma z_j}{a} \frac{\partial \pi_i^D}{\partial z_i} > 0 \quad \forall \ i \neq j,$$
(A.7)

which can be signed from (A.6). Finally, using $p_i^h = \gamma z_i$ from (4a) gives

$$\frac{\partial H_i}{\partial t_i} = -p_i^{h*} \frac{\gamma}{a} z_i^2 - 2N_i \gamma z_i < 0.$$
(A.8)

Substituting (A.7)–(A.8) into (A.4) yields the signs in eq. (18) in the main text.

A3. Proof of Proposition 2

We confine our analysis to the governments' optimal tax problem in Regime II. Substituting (16), (20) and (22), along with F_i in (23) into (24), the government's first-order condition can be rewritten as

$$\frac{\partial W_i}{\partial t_i} = \frac{N_i\Omega}{(1+2t_i)^2} \left[\gamma\Omega - \frac{4t_i\gamma\Omega}{(1+2t_i)} + 2\delta sd \right] + \frac{\gamma\Omega^2}{3a(1+2t_i)^2} \left[p_0^l sd + \frac{\delta sd\Omega}{(1+2t_i)} - \frac{\gamma t\Omega^2}{(1+2t_i)^2} \right]. \tag{A.9}$$

To show the concavity of W_i in t_i we establish the conditions under which W_i is rising in t_i at $t_i = 0$, but falling in t_i when $t_i \to \infty$. Evaluating (A.9) at $t_i = 0$ gives

$$\frac{\partial W_i}{\partial t_i}\Big|_{t=0} = \Omega \left[N_i (\gamma \Omega + 2\delta s d) + \frac{\gamma \Omega s d}{3a} (p_0^l + \delta \Omega) \right], \tag{A.10}$$

which is always positive for $\delta \geq 0$.

Evaluating (A.9) at $t_i \to \infty$, using L'Hôpital's rule and inserting the migration elasticity (10) gives

$$\frac{\partial W_i}{\partial t_i}\Big|_{t\to\infty} = \frac{\Omega N_i}{(1+2t_i)^2} \left[-\gamma \Omega \left(1 - \frac{\varepsilon p_0^l s d}{3}\right) + 2\delta s d\right].$$
 (A.11)

This is negative if the term in squared brackets is negative. The condition for this to hold is summarized in Proposition 2.

Further, W_i must be continuous in t_i because all terms in (A.9) are continuous functions of t_i . Therefore, if the condition for (A.11) to be negative is met, then W_i must be strictly concave in t_i and an interior optimum with $0 < t^* < \infty$ must exist. \Box

A4. Proof of Proposition 3

The proof is confined to showing that a negative fiscal value of a manager $(F_i < 0)$ is sufficient for $\partial W_i/\partial t_j < 0$ to hold in Regime I. Rearranging the countries' first-order condition for the optimal bonus tax (24), multiplying through by $(\partial z_i/\partial t_j)/(\partial z_i/\partial t_i)$ and using symmetry gives

$$[2\gamma z_i t_i - \delta sd] N_i \frac{\partial z_i}{\partial t_j} = -N_i z_i^2 \gamma \frac{\partial z_i / \partial t_j}{\partial z_i / \partial t_i} - F_i \frac{\gamma z_i}{a} \left(1 - \frac{\partial z_i / \partial t_j}{\partial z_i / \partial t_i}\right).$$
(A.12)

Substituting (A.12) in (26) gives, after cancelling terms

$$\frac{\partial W_i}{\partial t_j} = F_i \frac{\gamma z_i}{a} \frac{\partial z_i}{\partial t_j} \left(\frac{\partial z_i / \partial t_j}{\partial z_i / \partial t_i} - \frac{\partial z_i / \partial t_i}{\partial z_i / \partial t_j} \right) - N_i z_i^2 \gamma \frac{\partial z_i / \partial t_j}{\partial z_i / \partial t_i}.$$
 (A.13)

Since dz_i/dt_i and dz_i/dt_j are both negative and since $|dz_i/dt_i| > |dz_i/dt_j|$ follows from the stability of the Nash equilibrium, the first term in (A.13) has the same sign as F_i , whereas the second term in (A.13) is always negative. Hence $F_i > 0$ is a necessary, but not a sufficient condition for $dW_i/dt_j > 0$ [see Proposition 3(*i*)]. In contrast, $F_i < 0$ is a sufficient condition for $dW_i/dt_j < 0$ to hold, as stated in Proposition 3(*ii*). \Box

A5. Asymmetries between countries

Differentiating (24) with respect to s_A gives

$$\frac{\partial^{2}W_{i}}{\partial t_{i}\partial s_{i}} = \left[z_{i}^{2}\gamma + 2\gamma z_{i}t_{i}\frac{\partial z_{i}}{\partial t_{i}} - \delta s_{i}d\frac{\partial z_{i}}{\partial t_{i}}\right]\frac{\gamma z_{i}}{a}\left[\frac{\partial z_{i}}{\partial s_{i}} - \frac{\partial z_{j}}{\partial s_{i}}\right] \\
+ N_{i}\left[2z_{i}\gamma\frac{\partial z_{i}}{\partial s_{i}} + 2\gamma t_{i}\frac{\partial z_{i}}{\partial t_{i}}\frac{\partial z_{i}}{\partial s_{i}} - \delta d\frac{\partial z_{i}}{\partial t_{i}} + 2\gamma z_{i}t_{i}\frac{\partial^{2} z_{i}}{\partial t_{i}\partial s_{i}} - \delta s_{i}d\frac{\partial^{2} z_{i}}{\partial t_{i}\partial s_{i}}\right] \\
+ F_{i}\frac{\gamma}{a}\left[\frac{\partial z_{i}}{\partial s_{i}}\frac{\partial z_{i}}{\partial t_{i}} - \frac{\partial z_{j}}{\partial s_{i}}\frac{\partial z_{j}}{\partial t_{i}} + z_{i}\left(\frac{\partial^{2} z_{i}}{\partial t_{i}\partial s_{i}} - \frac{\partial^{2} z_{j}}{\partial t_{i}\partial s_{i}}\right)\right] \\
+ \frac{\gamma z_{i}}{a}\left[\frac{\partial z_{i}}{\partial t_{i}} - \frac{\partial z_{j}}{\partial t_{i}}\right]\left[t_{i}z_{i}\gamma\frac{\partial z_{i}}{\partial s_{i}} + p^{h}t_{i}\frac{\partial z_{i}}{\partial s_{i}} - \delta s_{i}d\frac{\partial z_{i}}{\partial s_{i}} - p^{l}d\right] \quad (A.14)$$

Our focus is on the boldfaced terms in the first, second and fourth lines of this equation. These terms all have the same sign as δ when $\partial z_i/\partial s_i > 0$ holds in (32), because $\partial z_i/\partial t_i < 0$ holds from (A.4) and $\partial^2 z_i/\partial t_i \partial s_i < 0$ holds from (A.8) and $\partial z_i/\partial s_i > 0$. Hence the boldfaced terms tend to cause an increase in t_i following a rise in s_i when $\delta > 0$, but lead to a decrease in t_i when $\delta < 0$.

References

- Acharya, V. (2003). Is the international convergence of capital adequacy regulation desirable? *Journal of Finance* 58, 2745-2781.
- Acharya, V., Pagano, M., Volpin, P. (2016). Seeking alpha: Excess risk taking and competition for managerial talent. *Review of Financial Studies*, 2565-2599.
- Adams, R.B. (2012). Governance and the financial crisis. International Review of Finance 12, 7-38.
- Bierbrauer, F., Brett, C., Weymark, J. (2013). Strategic nonlinear income tax competition with perfect labor mobility. *Games and Economic Behavior 82*, 292-311.
- Bannier, C., Feess, E., Packham, N. (2013). Competition, bonuses, and risk-taking in the banking industry. *Review of Finance 17*, 653-690.
- Barth, J., Lee, C., Phumiwasana, T. (2006). Deposit insurance schemes. In: Ch. Lee and A. Lee (eds.), *Encyclopedia of Finance*, 299-306.
- Bell, B., Van Reenen, J. (2014). Bankers and their bonuses. *Economic Journal 124*, 1-21.
- Besley, T., Ghatak, M. (2013). Bailouts and the optimal taxation of bonus pay. American Economic Review Papers & Proceedings 103, 163-167.
- Bhagat, S., Bolton, B. (2014). Financial crisis and bank executive incentive compensation. *Journal of Corporate Finance* 25, 313-341.
- Cheng, I.-H., Hong, H., Scheinkman, J.A. (2015). Yesterday's heroes. Compensation and risk at financial firms. *Journal of Finance 70*, 839-879.
- Dell'Ariccia, G., Marquez, R. (2006). Competition among regulators and credit market integration. Journal of Financial Economics 79, 401-430.
- Diamond, D., Dybvig, P. (1983). Bank runs, deposit insurance, and liquidity. Journal of Political Economy 91, 401-419.
- Dietl, H. M., Grossmann, M., Lang, M., Wey, S. (2013). Incentive effects of bonus taxes in a principal-agent model. *Journal of Economic Behavior and Organization 89*, 93-104.

- Efing, M., Hau, H., Kampkötter, P., Steinbrecher, J. (2015). Incentive pay and bank risk-taking: Evidence from Austrian, German, and Swiss banks. *Journal of International Economics 96*, 123-S140.
- Ehrlich, M.v., Radulescu, D. (2017). The taxation of bonuses and its effect on executive compensation and risk-taking: Evidence from the UK experience. *Journal* of Economics and Management Strategy, forthcoming.
- European Banking Authority (2016). Benchmarking of remuneration practices at the European Union level and data on high earners. 30 March 2016.
- Greve, P., Biemann, T., Ruigrok, W. (2015). Foreign executive appointments: A multilevel examination. Journal of World Business 50, 674-686.
- Greve, P., Nielsen, S., Ruigroka, W. (2009). Transcending borders with international top management teams: A study of European financial multinational corporations. *European Management Journal* 27, 213-224.
- Hakenes, H., Schnabel, I. (2014). Bank bonuses and bailouts. Journal of Money, Credit and Banking 46, 259-288.
- Haufler, A., Maier, U. (2016). Regulatory competition in capital standards with selection effects among banks. CESifo Working Paper No. 5839, Munich.
- Ho, P.-H., Huang, C.-W., Lin, C.-Y., Yen, J.-F. (2016). CEO overconfidence and financial crisis. Evidence from bank lending and leverage. *Journal of Financial Economics 120*, 194-209.
- International Monetary Fund (2010). A fair and substantial contribution by the financial sector. Final report for the G-20. Washington, D.C.
- Keen, M. (2011). Rethinking the taxation of the financial sector. *CESifo Economic* Studies 57, 1-24.
- Keen, M., Konrad, K. (2013). The theory of international tax competition and coordination. Handbook of Public Economics, Volume 5, 257-328.
- Kleven, H. Landais, C., Saez, E., Schultz, E. (2014). Migration and wage effects of taxing top earners. Evidence from the foreigners' tax scheme in Denmark. *Quarterly Journal of Economics 129*, 333-378.

- Laeven, L., Levine, R. (2009). Bank governance, regulation and risk taking. Journal of Financial Economics 93, 259-275.
- Lehmann, E., Simula, L., Trannoy, A. (2014). Tax me if you can! Optimal nonlinear income tax between competing governments. *Quarterly Journal of Eco*nomics 129, 1995-2030.
- Lipatov, V., Weichenrieder, A. (2015). Welfare and labor supply implications of tax competition for mobile labor. Social Choice and Welfare 45, 457-477.
- Markusen, J.R., Morey, E., Olewiler, N. (1995). Competition in regional environmental policies when plant locations are endogenous. *Journal of Public Economics* 56, 55-77.
- Philippon, T., Reshef, A. (2012). Wages and human capital in the U.S. finance industry: 1909-2006. Quarterly Journal of Economics 127, 1551-1609.
- Piketty, T., Saez, E., Stantcheva, S. (2014). Optimal taxation of top labor incomes: A tale of three elasticities. American Economic Journal: Economic Policy 6, 230-271.
- Radulescu, D. (2012). The effects of a bonus tax on manager compensation and welfare. *Finanzarchiv/Public Finance Analysis 68*, 1-16.
- Simula, L., Trannoy, A. (2010). Optimal income tax under the threat of migration by top-income earners. *Journal of Public Economics* 94, 163-173.
- Sinn, H.W. (1997). The selection principle and market failure in systems competition. Journal of Public Economics 66, 247-274.
- Sinn, H.W. (2003). Risktaking, limited liability, and the competition of bank regulators. *Finanzarchiv/Public Finance Analysis 59*, 305-329.
- Staples, C.L. (2008). Cross-border acquisitions and board globalization in the world's largest TNCs, 1995-2005. Sociological Quarterly 49, 31-51.
- Thanassoulis, J. (2012). The case for intervening in bankers' pay. The Journal of Finance 67, 849-895.

van Veen, K., Sahib, P.R., Aangeenbrug, E. (2014). Where do international board members come from? Country-level antecedents of international board member selection in European boards. *International Business Review 23*, 407-417.